Author's response to reviews

Title: Trends in all cause and liver-related hospitalizations in people with hepatitis B or C: a population-based linkage study

Authors:

Heather F Gidding (hgidding@nchecr.unsw.edu.au)
Gregory J Dore (gdore@nchecr.unsw.edu.au)
Janaki Amin (jamin@nchecr.unsw.edu.au)
Matthew G Law (mlaw@nchecr.unsw.edu.au)

Version: 4 Date: 6 January 2011

Author's response to reviews: see over
6 January 2011

To: Professor M Norton  
Editor-in-Chief, *BMC Public Health*

Re: Proof submission of manuscript 1949218944211241

Dear Professor Norton,

We have made all of the suggested editorial changes to our manuscript entitled:

“*Trends in all cause and viral liver-related hospitalizations in people with hepatitis B or C: a population-based linkage study*”

written by HF Gidding, GJ Dore, J Amin and MG Law.

Enclosed is our revised version with updated reference numbering. The manuscript now also includes the tables.

Yours sincerely

Heather Gidding  
Phone: +61 2 9385 0965  
Email: hgidding@nchecr.unsw.edu.au